Author's response to reviews

Title: Pulmonary immune responses to 2009 pandemic influenza A (H1N1) virus in mice

Authors:

Jin Lv (lvjin6630@hotmail.com)
Dan Wang (wdyy_618@sina.com)
Yanhong Hua (huayanhong413@gmail.com)
Shijia Pei (peishijia@163.com)
Jin Wang (wjninr1981@163.com)
Rui He (hrututu@hotmail.com)
Ye Wu (wuye2004@hotmail.com)
Xiliang Wang (snss@sina.com)
Na Jia (jiana79_41@hotmail.com)
Qisheng Jiang (jqs598@sina.com)

Version: 2
Date: 16 December 2013

Author's response to reviews:

In this letter, we have made some changes such as: list email addresses for all authors in the title page, provide competing interests declaration, and clarify contributions of all authors in the end of the paper.